Author's response to reviews

Title: Prospective, open, multi-centre phase I/II trial to assess the safety and efficacy of neoadjuvant radiochemotherapy with docetaxel and oxaliplatin in patients with adenocarcinoma of the oesophagogastric junction

Authors:

Markus Moehler (moehler@mail.uni-mainz.de)
Ines Gockel (ines.gockel@unimedizin-mainz.de)
P. Roessler (leitung@radioonkologie.klinik.uni-mainz.de)
Dirk Arnold (darnold@uke.uni-hamburg.de)
Tanja Trarbach (Tanja_Trarbach@uk-essen.de)
G. Klautke (strahlentherapie@med.uni-rostock.de)
C. Roedel (ClausMichael.Roedel@kgu.de)
Baruch Brenner (brennerb@clalit.org.il)
Thomas Thomaidis (Thomas.Thomaidis@unimedizin-mainz.de)
Heinz Schmidberger (heinz.schmidberger@unimedizin-mainz.de)
Hauke Lang (hauke.lang@unimedizin-mainz.de)
Peter R. Galle (galle@1-med.klinik.uni-mainz.de)
Carl C. Schimanski (christoph.schimanski@unimedizin-mainz.de)

Version: 6 Date: 30 January 2013

Author's response to reviews:

Dear Sirs,

as wished I included the section

Authors’ contributions

MM, IG, BB, CCS, and HS participated in the design of the study and performed the statistical analysis. MM, PR, DA, TTh coordinated and helped to draft the manuscript. All authors participated actively in the patient recruitment, read and approved the final manuscript.

Thank you for further processing.

Best regards

MM